Author's response to reviews

Title: High Magnitude of under nutrition among HIV infected adults who have not started ART in Tanzania--A call to include nutrition care and treatment in the test and treat model

Authors:

Bruno Sunguya (Sunguya@gmail.com)

Nzovu Ulenga (nulenga@mdh-tz.org)

Hellen Siril (neemasiril@gmail.com)

Sarah Puryear (sarah.puryear@gmail.com)

Eric Aris (aeric@mdh-tz.org)

Expedito Mtisi (emtisi@gmail.com)

Edith Tarimo (etarimo54@yahoo.co.uk)

David Urassa (durassa2@yahoo.com)

Wafaie Fawzi (mina@hsph.harvard.edu)

Ferdnand Mugusi (fm.mugusi@gmail.com)

Version: 3 Date: 04 Jul 2017

Author’s response to reviews:

1. In the Ethics and consent to participate section please include the trial registration number of the parent study (TOV3).

The trial registration number is now inserted in page 15

2. In the declarations section Consent for publication refers to consent from the patients to personally identifying data published. As your manuscript does not contain identifying data from any individual person, please state “Not applicable” in this section. But please incorporate the statement on consent for you to use the data into the methods section.

The revision is now made according to this recommendation

3. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).
Currently, the contributions of some of the authors do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. We recommend that that you adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; and

- been involved in drafting the manuscript or revising it critically for important intellectual content; and

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

Please also include a statement in the Authors' contributions section to clarify whether all authors have read and approved the final version of the manuscript.

Revisions made according to the comment in page 9 line 23-28.

4. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided.

Abbreviation list has been made according to the comment. Page 9-10

5. On the title page please include email addresses for all listed authors.

List of emails for all authors has been made in Page 1

6. Please change the heading “conclusions and recommendations” to “conclusions” in both the abstract and the main text.

The change has been made in both abstract and main text.